Author's response to reviews

Title: A Systematic Review of Evidence-based Treatment for Depersonalization-derealization Disorder (DPRD)

Authors:

Taryn Amos-Williams (tarynamos@gmail.com)
Eli Somer (somer@research.haifa.ac.il)
Dan J Stein (Dan.Stein@uct.ac.za)

Version: 2 Date: 22 January 2013

Author's response to reviews:

Dear Editors

Depersonalization-derealization disorder (DPRD) is a distressing and impairing condition that is not well understood. Nevertheless, given the growing interest in its pathogenesis, a systematic review of controlled pharmacotherapy and psychotherapy trials is timely. To date, however, no systematic review has been undertaken in this area. We have undertaken the first systematic review of randomized controlled trials of DPRD.

The scope of BMC Psychology includes aspects encompassing human behavior and the mind, including developmental, clinical, cognitive, experimental, social, evolutionary and educational psychology, as well as personality and individual differences. We think this paper falls within this scope, and constitutes a useful contribution to the field.

We therefore do hope that you will consider this paper for publication in your journal.

Regards

Prof. Eli Somer, Ph.D., Senior Clinical Psychologist
School of Social Work, University of Haifa, Haifa 31905, Israel
Somer Psychotherapy and Research www.somer.co.il
Work: +972-4-8360494
Mobile: +972-54-4659525
E-mail: eli.somer@me.com

Professor Dan J Stein, FRCPC, PhD, DPhil
Head: Department of Psychiatry and Mental Health, UCT & GSH
Groote Schuur Hospital (J-2), Anzio Rd, Observatory 7925, Cape Town, South Africa